Title: Alterations of regional homogeneity in pediatric bipolar depression: A resting-state fMRI study

Authors:

Weijia Gao (gaoweijia0425@163.com)
Qing Jiao (bingbao17@163.com)
Shaojia Lu (ronaldo0319@163.com)
Yuan Zhong (118202038@qq.com)
Rongfeng Qi (fmrigirf@126.com)
Dali Lu (liulang7026971@163.com)
Qian Xiao (xiaqian851112@126.com)
Fan Yang (fanyang79@gmail.com)
Guangming Lu (cjr.luguangming@vip.163.com)
Linyan Su (gaoweijia0425@sina.com)

Version: 4 Date: 11 July 2014

Author's response to reviews:

Dear Mr. Carlo Rye Chua,

We thank you for providing us the opportunity to revise our manuscript titled: “Alterations of regional homogeneity in pediatric bipolar depression: A resting-state fMRI study”. We also greatly appreciate the reviewer’s comments and suggestions, which have helped to improve the quality of our manuscript. According to the reviewer’s suggestion, we have put the x MNI coordinate into Figure 2.

Thank you for your hard work again and we hope you will be satisfied with our replies. We are ready to deal with the questions if you have any. Many thanks!

Yours sincerely,

Linyan Su

Mental Health Institute, the Second Xiangya Hospital, Central South University, Changsha, China